Accelerator Status Report

Run Schedule: March 11, 2010 to July 22, 2010
June 2, 2010
Injector    Energy 56.62 MeV
· (6/1) Reza optimized the injector for all three beams.

· Injector needs optimization before Hall A is back on-line.

· (5/23) Cavity #8 of the quarter cryo module malfunctioned. The control module was replaced. 

· Since Hall B was the only hall taking beam for the last few days sometimes Hall B laser needed to be adjusted to compensate for drifts.  

Accelerator   Linac Energy 503.25 MeV
· The accelerator has been running well.
· RF separators trip off sometimes. EES is investigating. 
· (5/27) CHL tripped off. Do not know the cause yet. 

· Some random RF issues
· BSY dump window still has problems. But we can run beam there if the presure recovers.
· The accelerator was running well and the optics stable.

· (5/25) BSY dump window re-pressurized. We should be able to run beam there without problems.

· (5/25) Zone 2L17 LCW leaked over the weekend. The leaking hose was replaced.
· (5/24) RSEP9A mismatched and was repaired.
Hall A---Pass 1, Prex from March 11 to May 24, 2010
· Hall A has been down and is planning to come back online tonight.

· Hall A stopped taking beam on May 20 due to its vacuum issues. 
Hall B ---Pass 5, g9-Frost from March 11 to July 22, 2010
· No problems with Hall B beam.

· (5/25) Hall B beam was not stable after yesterday’s beam study. Five hours was lost.

· No issues with Hall B beam. 

      Once in a while Hall B laser phase needed to be adjusted a degree or two to compensate

            for drifts since Hall A did not take beam.
Hall C---QWeak from June 8 to August 15, 2010
· QWeak start-up will be delayed again for a few days. Not sure when yet.

· Hall C is planning to try beam on June 8.

· QWeak installation
